Craig Chambers CSE 501 Reading Assignments

Reading Assignment, Weeks #5, #6

For: Read:

2/1 Review section 18.2 2/5 Section 15.4, skim section 15.6, [Dean & Chambers 94]

2/8 [Callahan et al. 86], [Consel & Danvy 93] 2/10 Section 17.5 2/12 [Wilson & Lam 95], [Steensgaard 96]

Bibliography:

[Dean & Chambers 94] Jeff Dean and Craig Chambers. Towards Better Inlining Decisions Using Inlining Trials. In L&FP’94.

[Callahan et al. 86] David Callahan, Keith D. Cooper, Ken Kennedy, and Linda Torczon.

Interprocedural Constant Propagation. In PLDI ’86. [Consel & Danvy 93] Charles Consel and Olivier Danvy. Partial Evaluation: Principles and Perspectives. Tutorial notes included in POPL ’93.

[Wilson & Lam 95] Robert P. Wilson and Monica S. Lam. Efficient Context-Sensitive Pointer Analysis for C Programs. In PLDI ’95.

[Steensgaard 96] Bjarne Steensgaard. Points-to Analysis in Almost Linear Time. In POPL’96.